Author's response to reviews

Title: Incidence and risk factors for mortality in HIV-1-infected patients in China: a cross-sectional study

Authors:

Hui H Zheng (zhenghui8946@hotmail.com)
Lu L Wang (wanglu_bj2013@126.com)
Qing Q Wang (wqnimu@163.com)
Peng P Huang (hp19880310@163.com)
Wei W Guo (gwei_bj@sohu.com)
Zhihang Z Peng (zhihangpeng@njmu.edu.cn)
Rongbin R Yu (rongbinyu@njmu.edu.cn)
Ning N Wang (wangning_12@126.com)

Version: 2 Date: 24 February 2014

Author's response to reviews: see over
Dear Editor:

We would like to submit the enclosed manuscript entitled “Incidence and risk factors for mortality in HIV-1-infected patients in China: a cross-sectional study”, which we wish to be considered for publication in “BMC public health”. No conflict of interest exits in the submission of this manuscript, and manuscript is approved by all authors for publication. I would like to declare on behalf of my co-authors that the work described was original research that has not been published previously, and not under consideration for publication elsewhere, in whole or in part. All the authors listed have approved the manuscript that is enclosed.

In this work, we evaluated a cross-sectional study based on the whole Chinese death people after HIV infection from 2006 to 2010. Sample size is large, compared to previous research. The result of this study will be better to provide the basis for the development of public health strategies. I hope this paper is suitable for “BMC public health”.

We deeply appreciate your consideration of our manuscript, and we look forward to receiving comments from the reviewers. If you have any queries, please don’t hesitate to contact me at the address below.

Thank you and best regards.

Yours sincerely,

Zhihang Peng, Ph.D.
Department of Epidemiology and Biostatistics, School of Public Health, Nanjing Medical University, Nanjing 211166, China
E-mail: zhihangpeng@njmu.edu.cn